\documentclass[12pt]{article}

\usepackage{graphics}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}

\begin{document}

\textbf{Math 365: Honors Analysis I}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 2}\quad \textsl{September 29, 2000}

\bigskip
1. Find two independent power series solutions and their radii of
convergence for the equation
\[ y'' + x y' + y = 0 \]

\newpage

2.  Find two independent Frobenius series solutions for the 
equation
\[ 2x^2 y'' - 3x y' + 3(x+1) y = 0 \]


\end{document}
